Author’s response to reviews

Title: Sexual behaviours and determinant factors among students at Bahir Dar University. A cross sectional study

Authors:

Wondemagegn Lingerew Mulu (Wondem_32@yahoo.com)
Bayeh Beyene Abera (bayeabera15@gmail.com)
Mulat Melaku Yimer (yimermulat@gmail.com)

Version: 2 Date: 2 December 2013

Author's response to reviews: see over
AUTHOR'S COVER LETTER

Title:  Sexual behaviours and determinant factors among students at Bahir Dar University: A cross sectional study

Editor-in-Chief
BioMed Central  BMC Reproductive Health

Dear Dr,

We hereby submit a manuscript entitled with: Sexual behaviours and determinant factors among students at Bahir Dar University for possible consideration of publication in BMC Reproductive Health. We strongly feel that the manuscript is original and has not been submitted anywhere else for publication. Moreover, the author(s) declare that they have no competing interest.